Author's response to reviews

Title: An evaluation of alcohol attendances to an inner city emergency department before and after the introduction of the UK Licensing Act 2003.

Authors:

Andrew J Durnford (axd473@bham.ac.uk)
Tommy J Perkins (txp464@bham.ac.uk)
James M Perry (j.perry@doctors.org.uk)

Version: 6 Date: 30 October 2008

Author's response to reviews: see over
RE: An evaluation of alcohol attendances to an inner city emergency department before and after the introduction of the UK Licensing Act 2003.

MS: 9581434751818353 Andrew J Durnford, Tommy J Perkins and James M Perry

Dear Dr Graham,

I am pleased to submit our final manuscript with the requested formatting changes. We look forward to hearing from you regarding the press release.

Yours sincerely,

Dr Tommy Perkins